B\'ezier is B\'ezier.

Here is a matrix as displayed equation:
\[
	\left[
	\begin{tabular}{ccc}
	1 & 0 & 0\\
	0 & 1 & 0\\
	0 & 0 & 1\\
	\end{tabular}
	\right]
	=
	\left[a~b~c\right]
	\left[\begin{tabular}{c}x\\y\\z\end{tabular}\right]
\]
The left and right have to match.  Use right. to get a one-sided symbol:
\[
	F(t) = \left\{
	\begin{tabular}{ll}
	t,&\mbox{$0<t<1$}\\
	0,&\mbox{otherwise}
	\end{tabular}
	\right.
\]
	

Here is how to subscript $V_1$ and superscript $V^2$.  Put things
in dollar signs to get $u$ and $v$.

Often you will have derivations that look like this:
\begin{eqnarray*}
F &=& XY + XZ \\
  &=& X(Y+Z)
\end{eqnarray*}

Here's how to do a figure.
\begin{figure}
\begin{center}
% Do not include the .eps, .ps, or .pdf extension on the figure file name
% The basic use of include graphics is:
%\includegraphics{file}

% If you want to specify the height, use the optional height argument.
%\includegraphics[height=2in]{file}

\end{center}

\caption[Test figure short caption]{Test figure with a really, really, really,
	really, really, really, really, really, really, really, really, really, really,
	really, really, long caption.}
\label{fig:test}
\end{figure}
Here's how you refer to Figure~\ref{fig:test}.  You should also uncomment
out the includegraphics, and the ``file'' is the name of the figure file
(do not put an extension on the file).

To get an equation with an equation number, use begin\{equation\} like
this:
\begin{equation}
F = ma
\label{eq:force}
\end{equation}
and refer to it as Equation~\ref{eq:force}.
